Author's response to reviews

Title: Reliability and validity of the Thai version of the PHQ-9

Authors:

Manote Lotrakul (ramlt@mahidol.ac.th)
Sutida Sumrithe (rasda@mahidol.ac.th)
Ratana Saipanish (rarsp@mahidol.ac.th)

Version: 5 Date: 20 May 2008

Author's response to reviews:

We have made changes in the manuscripts according to the requests of the editorial team:

1) Affiliations - Please write each affiliation on a new line.
Changes done accordingly

2) Each email address should be on a new line and should be preceded by the author's initials, not the full name.
Changes done accordingly

3) Table and figure placement text Please remove this.
Changes done accordingly

4) Abbreviations - Please present these as a paragraph rather than a list.
Changes done accordingly

5) Please title 'Figure legends' as such.
Changes done accordingly